Author's response to reviews

Title: Prospective, randomized evaluation of a personal digital assistant-based research tool in the emergency department

Authors:

Morris L Rivera (moispo1@hotmail.com)
Jason Donnelly (jason_donnelly@nymc.edu)
Blair A Parry (bparry@partners.org)
Anthony Dinizio (dinizio@bc.edu)
Charles L Johnson (charlie@studymaker.com)
Jeffrey A Kline (jkline@carolina.rr.com)
Christopher Kabrhel (ckabrhel@partners.org)

Version: 4 Date: 2 January 2008

Author's response to reviews:

As requested we have revised the author contribution paragraph to more specifically address each of the authors' involvement in the study. We have also revised the formatting of the references to include the names of all relevant authors.